Author’s response to reviews

Title: Factors influencing microbial colonies in the air of operating rooms

Authors:
Ling Fu Shaw (lfu@vghtpe.gov.tw)
Ian Horng Chen (a0746@tpech.gov.tw)
Chii Shya Chen (a0745@tpech.gov.tw)
Hui Hsin Wu (anatom1963@gmail.com)
Li Shing Lai (g49007007@yahoo.com.tw)
Yin Yin Chen (g49007007@gmail.com)
Fu Der Wang (fdwang@vghtp.gov.tw)

Version: 1 Date: 02 May 2017

Author’s response to reviews:
Thank you for the editor's comments on our manuscript entitled “Factors influencing microbial colonies in the air of operating rooms.” (Manuscript No. INFD-D-17-00383). The manuscript has been revised in accordance with the comments and then edited English language by AJE.

Sincerely,
Yin Yin Chen